



A Comprehensive Study of ChatGPT: Advancements, Limitations, and Ethical Considerations in Natural Language Processing and Cybersecurity

Moatsum Alawida ^{1,*}, Sami Mejri ², Abid Mehmood ¹, Belkacem Chikhaoui ^{3,*} and Oludare Isaac Abiodun ⁴

- ¹ Department of Computer Sciences, Abu Dhabi University, Abu Dhabi P.O. Box 59911, United Arab Emirates; abid.mehmood@adu.ac.ae
- ² Center for Teaching and Learning, Khalifa University of Science and Technology, Abu Dhabi P.O. Box 127788, United Arab Emirates; sami.mejri@ku.ac.ae
- ³ Applied Artificial Intelligence Institute, TELUQ University, Montreal, QC H2S 3L5, Canada
- ⁴ Department of Computer Science, University of Abuja, Gwagwalada 900110, Nigeria; oludare.abiodun@uniabuja.edu.ng
- * Correspondence: moatsum.alawida@adu.ac.ae (M.A.); belkacem.chikhaoui@teluq.ca (B.C.)

Abstract: This paper presents an in-depth study of ChatGPT, a state-of-the-art language model that is revolutionizing generative text. We provide a comprehensive analysis of its architecture, training data, and evaluation metrics and explore its advancements and enhancements over time. Additionally, we examine the capabilities and limitations of ChatGPT in natural language processing (NLP) tasks, including language translation, text summarization, and dialogue generation. Furthermore, we compare ChatGPT to other language generation models and discuss its applicability in various tasks. Our study also addresses the ethical and privacy considerations associated with ChatGPT and provides insights into mitigation strategies. Moreover, we investigate the role of ChatGPT in cyberattacks, highlighting potential security risks. Lastly, we showcase the diverse applications of ChatGPT in different industries and evaluate its performance across languages and domains. This paper offers a comprehensive exploration of ChatGPT's impact on the NLP field.

Keywords: ChatGPT; language generation; natural language processing (NLP); cybersecurity; ChatGPT applications

1. Introduction

ChatGPT is a sophisticated chatbot developed with OpenAI. It is based on the GPT (Generative Pre-Trained Transformer) language model technology and is capable of understanding and interpreting user requests and generating appropriate responses in natural human language, allowing it to complete a wide range of text-based tasks such as answering questions and generating responses. It is a significant innovation in NLP and artificial intelligence. OpenAI, a research laboratory founded in 2015, has gained significant support and investment, allowing it to make rapid progress in the development of AI technologies, including other products such as DALL-E. GPT, the technology on which ChatGPT is based, is a language model that uses a combination of generative and discriminative techniques to produce responses indistinguishable from natural human language, by learning from vast amounts of data, including the entire internet. This technology has the potential to greatly reduce the time required for research tasks and may potentially render traditional research authors obsolete [1].

ChatGPT has shown impressive results on various benchmark datasets and has been widely adopted in various natural languages processing applications such as chatbots, dialogue systems, and text generation. The remarkable performance of ChatGPT has sparked a renewed interest in the field of language models and their potential to revolutionize the way we interact with machines. Interestingly, ChatGPT is a powerful tool for various NLP



Citation: Alawida, M.; Mejri, S.; Mehmood, A.; Chikhaoui, B.; Isaac Abiodun, O. A Comprehensive Study of ChatGPT: Advancements, Limitations, and Ethical Considerations in Natural Language Processing and Cybersecurity. *Information* **2023**, *14*, 462. https:// doi.org/10.3390/info14080462

Academic Editors: Moutaz Alazab and Ammar Alazab

Received: 8 July 2023 Revised: 3 August 2023 Accepted: 13 August 2023 Published: 16 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). tasks, including language translation, question answering, text completion, and more [2]. ChatGPT's massive size and advanced training techniques have also allowed it to demonstrate a level of general knowledge and reasoning abilities that were previously unseen in language models [3,4].

In this study, we aim to provide a comprehensive overview of ChatGPT, its strengths, limitations, and future implications. We also explore the current state-of-the-art language models and compared them with ChatGPT. Therefore, this study will be valuable for researchers, practitioners, and anyone interested in cutting-edge developments in the field of natural language processing.

1.1. Motivation

The main motivation behind this study is to provide useful information on the latest knowledge of ChatGPT which researchers can explore to advance the state-of-the-art in language processing and natural language understanding [5]. Since ChatGPT is a powerful AI model that can perform a wide range of tasks with human-like proficiency, it means researchers can find a more advanced language model like ChatGPT to resolve problems and to implement projects. OpenAI's objective is to drive the progression of AI by facilitating the creation of novel applications in diverse domains, including NLP and beyond. Its overarching goal is to enhance the field of AI research and to empower researchers and developers with tools and resources to effectively explore and innovate in these areas. Therefore, the motivation for this study is also to demonstrate the potential of large language models and to encourage further investments in this area of AI research for economic development and sustainability.

1.2. Main Contribution of This Paper

The main contributions of this paper can be summarized as follows:

- 1. Comprehensive Analysis of ChatGPT: This paper conducts a thorough analysis of ChatGPT's capability to generate human-like text across diverse styles and topics. This analysis is based on the model's extensive training on large amounts of diverse text data. It examines the architecture, training data, and evaluation metrics of ChatGPT, providing a comprehensive understanding of its capabilities.
- 2. Exploration of ChatGPT's Significance and Implications: The paper offers insights into ChatGPT as an emerging language model for generative text and discusses its future implications. It provides background information on ChatGPT, highlighting its potential impact on natural language processing tasks and its suitability for various applications. Additionally, it examines the ethical considerations and privacy risks associated with large language models like ChatGPT and suggests ways to address them.
- 3. Comparative Analysis and Evaluation: The paper compares ChatGPT with other language generation models, assessing its strengths and limitations. It highlights ChatGPT's performance in natural language processing tasks such as language translation, text summarization, and dialogue generation. Furthermore, the paper explores the potential applications of ChatGPT in different industries and evaluates its performance across various languages and domains. Lastly, it examines the overall impact of ChatGPT on the field of natural language processing.

1.3. Paper Organization

The paper is structured in the following manner: Section 2 provides related work, and Section 3 provides an overview of ChatGPT, including its various types. Section 4 offers a comparative analysis of ChatGPT with other language generation models. Section 5 delves into the topic of dialogue generation with ChatGPT. Section 6 addresses privacy concerns related to ChatGPT. Section 7 investigates the various applications of ChatGPT in business and industry. Section 8 details the process of training and fine-tuning ChatGPT for specific tasks. Section 9 evaluates the language generation quality of ChatGPT. Section 10 assesses the performance of ChatGPT on different languages and domains. Section 11 provides the overview of ChatGPT in cybersecurity. Section 12 provides insight into the future of ChatGPT. Lastly, Section 13 concludes the paper.

2. Related Work

ChatGPT is a powerful language model that has been used for text generation in many different fields. Researchers have used ChatGPT to create text content that is relevant and coherent for specific industries [6], such as healthcare [7], tourism [6], and future development [8]. ChatGPT's versatility has led to innovative research and new applications in the real world [9,10].

In the paper [11], the authors highlighted some of the concerns in using large language models (LLMs) like ChatGPT. LLMs have the potential to revolutionize research methods, but there are also concerns about the quality, accuracy, and transparency of research outcomes that could be generated using LLMs. LLMs may contain errors, biases, and plagiarism. Additionally, LLMs may replicate and amplify the cognitive biases of the humans who train them. This means that there is a risk that LLMs could be used to generate inaccurate, biased, or plagiarized research.

The authors of the text provide guidance for researchers who use LLMs. They emphasize the importance of recognizing and mitigating the potential risks associated with using LLMs. This includes carefully validating the information generated with LLMs, ensuring accuracy, and meticulously assessing the sources behind the content. The authors conclude by stating that careful scrutiny is essential to preserve research integrity amidst the transformative capabilities of LLMs.

The authors explored the implications of using ChatGPT in the tourism industry [6], with a focus on striking a balance between convenience and challenges. The study's findings, based on interviews with professionals from various fields within the tourism industry, provide a comprehensive perspective. The integration of ChatGPT into the tourism industry presents a complex picture, with both convenience and challenges.

In the paper [8], the authors studied how ChatGPT is changing scientific research, including data processing, hypothesis generation, collaboration, and public outreach. They looked at potential problems and ethical questions when using ChatGPT in research, emphasizing the need to balance AI-driven innovation with human knowledge. The authors discussed ethical issues in various computing fields and how ChatGPT can pose challenges. They also pointed out biases and limitations of ChatGPT. Despite controversies, ChatGPT has gained attention in academia, research, and industry in a short time.

Another paper examined the accuracy of the content generated with ChatGPT in healthcare [7]. The results showed that most of the content was useful in some way, but that all statements provided should be checked more carefully. ChatGPT is a narrative LLM for medical personnel.

3. An Overview of ChatGPT

Three different versions of GPT, i.e., GPT-1, GPT-2, GPT-3, and GPT-4 were released by OpenAI in the past. The four versions of GPT differ in size. Each new version was trained by scaling up the data and parameters. Table 1 shows the different parameters used by the three versions of GPT and the different features each version supports [12,13].

ChatGPT was developed on top of the GPT-3 language model using reinforcement learning from human feedback (RLHF). The initial model was trained using supervised fine-tuning, where human AI trainers participated in conversations, playing both the role of the user and an AI assistant. The trainers were provided with model-generated suggestions to assist them in composing their responses. ChatGPT and InstructGPT are both variants of the GPT language model, but they have different focuses and training data. InstructGPT is specifically designed for generating instructional text and providing step-by-step guidance, while ChatGPT is a more general-purpose conversational AI model that can be used for a variety of text-based tasks.

	GPT-1	GPT-2	GPT-3	GPT-4
Parameters	117 million	1.5 billion	175 billion	300 billion
Decode Layers	12	48	96	128
Hidden Layers	768	1600	12,288	20,480
Context Token Size	512	1024	2048	4096
Fine Tuning Datasets	Limited	More	Many	Extensive
Fine Tuning Tasks	Few	More	Many	Extensive
Language Understanding	Limited	Improved	Advanced	Highly Advanced
Text Generation	Basic	Advanced	Very Advanced	Exceptional
Sentiment Analysis	Not Supported	Not Supported	Supported	Enhanced
Text Summarization	Not Supported	Not Supported	Supported	Enhanced
Text Correction	Not Supported	Not Supported	Supported	Enhanced

Table 1. Parameters used for training and features of GPT-1, GPT-2, GPT-3, and GPT-4.

A new dialogue dataset was created by combining the InstructGPT dataset, which was transformed into a dialogue format, with other dialogue data. This new dataset was used to fine-tune both ChatGPT and InstructGPT, allowing them to perform better in dialogue-based tasks. Overall, while both models are based on the GPT architecture, they have different strengths and use cases. InstructGPT is ideal for tasks that require providing detailed instructions, while ChatGPT is better suited for generating natural language responses in conversational settings.

ChatGPT uses reinforcement learning to fine-tune the model, and the data are collected for comparison, consisting of multiple model responses ranked by quality. The data are gathered from conversations between AI trainers and the chatbot. The trainer selects a randomly selected model-generated message, samples, and ranks during this process. Using reward models, it is fine-tuned by using Proximal Policy Optimization, and the process is repeated several times [14]. Figure 1 shows the basic model used to train ChatGPT, and the steps are listed as follows:



Figure 1. ChatGPT training model.

- The input layer takes in a sequence of words and converts them into numerical representations called embeddings. The embeddings are passed to the transformer layers.
- The transformer layers are made up of multi-head self-attention mechanisms and feed-forward neural networks. The self-attention mechanisms allow the model to focus on specific parts of the input when generating a response, while the feed-forward neural networks allow the model to learn and extract features from the input.

• The transformer layers are stacked on top of each other and are connected through residual connections. This allows the model to learn and extract features at different levels of abstraction.

The GPT models are built on Google's transformer architecture, which was introduced in 2017 through the paper "Attention is all you need" [15]. Within the field of natural language processing (NLP), researchers have explored various algorithms to address challenges in human language, including morphological variations, synonyms, homonyms, syntactic and semantic ambiguity, co-references, pronouns, negations, alignment, and intent. While earlier computational linguistics methods (such as grammatical, symbolic, statistical approaches, and early deep-learning models like LSTM) faced limitations in resolving these issues, transformers emerged as a successful solution.

Transformers, specifically through their advanced mechanism called multi-head selfattention, have demonstrated the ability to automatically detect and handle many linguistic phenomena mentioned above. This capability is a result of being exposed to vast amounts of human language data.

The original transformer architecture consists of an encoder and a decoder, which effectively handle complex sequence-to-sequence patterns involving both left- and right-side context sensitivity—common in natural language. Both the encoder and decoder comprise multiple feed-forward layers that employ the self-attention technique mentioned earlier. The number of these layers typically ranges from 12 to 24, depending on the model's complexity and can go up to 128 in the largest GPT-4 model. These layers have been proven to be able to capture various levels of linguistic ambiguity and complexity, encompassing punctuation, morphology, syntax, semantics, and more intricate interactions. Essentially, by being exposed to language data, these models swiftly learn the essential components of the standard NLP pipeline. ChatGPT can be used for a wide range of applications, as shown in Table 2.

Task	Description
Text Generation	ChatGPT can be used to generate a wide range of text, such as articles, essays, stories, and poetry. It can also be used to generate responses to user input in natural human language.
Question Answering	It can be used to answer questions, such as providing definitions, per- forming calculations, and providing information on a wide range of topics.
Content Creation	It can be used to generate content for websites, social media, and other platforms. It can also be used to generate product descriptions, reviews, and other types of content.
Language Translation	ChatGPT can be fine-tuned to perform language translation, translating text from one language to another.
Dialog Generation	ChatGPT can be used to generate responses in a conversational context, making it suitable for building chatbots, virtual assistants, and other conversational systems.
Text Summarization	ChatGPT can be fine-tuned to perform text summarization, condensing long text into shorter, more concise versions.
Sentiment Analysis	ChatGPT can be fine-tuned to perform sentiment analysis, analyzing text to determine the expressed sentiment (positive, negative, or neutral).
Text Completion	ChatGPT can be used to complete text. Given a partial text, it can predict the next word, sentence, or even a whole paragraph.
Text Correction	ChatGPT can be fine-tuned to perform text correction, rectifying grammar, spelling, and punctuation errors in the text.

Table 2. Applications of ChatGPT: a versatile language model for various text-related tasks [16].

These are just a few examples of the many possible applications of ChatGPT where it can be useful [1,16]. However, there are several limitations of ChatGPT, as given in Table 3.

Table 3. Limitations of ChatGPT.

Limitation	Description
Bias	Bias may be present in the train data, leading to unfair or inaccurate predictions
Lack of Contextual Understanding	Lack of ability to understand the context of the input, lead- ing to inaccurate prediction
Lack of Common Sense	Lack of common sense knowledge may limit the ability to understand and respond to certain types of questions
Require a large number of computa- tional resources	Requires a significant amount of computational resources to run making it difficult to deploy on some devices
Dependent on a Large amount of data	Requires a large amount of data to perform well on new tasks
Lack of interpretability	Difficult understanding how it makes predictions as it is based on neural networks

4. A Comparative Study of ChatGPT and Other Language Generation Models

Language generation models are an important area of research in the field of NLP. For example, ChatGPT can generate human-like text and perform a variety of NLP tasks. However, it is not the only model available for language generation. Other models such as GPT-2, GPT-3, BERT, RoBERTa, T5, XLNet, Megatron, and ALBERT have also been developed and have shown good performance in language generation and other NLP tasks [17]. Many studies have been conducted to compare the performance of various language models, including ChatGPT [18].

There are several research studies with an emphasis on language processing models and differences in the quality of outputs. A recent comparative study of GPT-2 and GPT-3 language models by researchers at the Indian Institute of Technology compared the performance of GPT-2 and GPT-3 [19]. The study looks at the quality of the language generated with the two models, as well as their ability to complete a variety of NLP tasks. The study concludes that GPT-3 outperforms GPT-2 in terms of language generation quality and task completion accuracy.

Another study that may be relevant is a comparative study of GPT-3 and BERT language models [20,21]. This study compared the performance of GPT-3 with BERT, which is another state-of-the-art language model developed by Google [22]. The study looked at the ability of the two models to complete a variety of natural language understanding tasks and concluded that GPT-3 outperforms BERT in many cases [23], but BERT is better in certain tasks such as named entity recognition [24]. Here are a few examples of tasks where GPT-3 has been found to outperform BERT in the scope of these relevant studies:

- Language generation: GPT-3 has been found to generate more fluent and naturalsounding language than BERT in several studies. This is likely due to GPT-3's larger model size and training data, which allows it to capture more nuanced relationships between words and phrases [23].
- Question answering: GPT-3 has been found to be more accurate than BERT in answering questions based on a given context. This is likely due to GPT-3's ability to generate text, which allows it to provide more detailed and informative answers [25].
- Text generation: GPT-3 has been found to generate more coherent and coherently written text than BERT in several studies. This is likely due to GPT-3's ability to generate text, which allows it to generate more complete and well-formed sentences [26].
- Text completion: GPT-3 has been found to be more accurate than BERT in completing the text, especially in the case of long-form text such as articles and essays [27].

- Summarization: GPT-3 has been found to generate more fluent and informative summaries than BERT in several studies. This is likely due to GPT-3's ability to understand and analyze the content of a text, which allows it to generate more accurate and informative summaries [13,28].
- Sentiment analysis: GPT-3 has been found to be more accurate than BERT in determining the sentiment of text, such as whether the text expresses a positive, negative, or neutral sentiment [29].
- Text classification: GPT-3 has been found to be more accurate than BERT in classifying text into different categories, such as news articles, social media posts, and customer reviews [30].
- Dialogue systems: GPT-3 has been found to be more accurate than BERT in generating natural and coherent responses in dialogue systems such as chatbots [31].

It is important to note that while GPT-3 has been found to outperform BERT in many cases, BERT is better in certain tasks such as named entity recognition, which is a task of identifying and classifying named entities in a given text [32].

Table 4 provides a comparison of popular language models based on various criteria such as model name, training data, model size, structure, performance, advantages, and disadvantages. The table includes information on models such as GPT-2, GPT-3, GPT-4, BERT, RoBERTa, T5, XLNet, Megatron, and ALBERT. It shows the amount of data used to train the model and the number of parameters, the structure of the model, and its performance on certain tasks. It also highlights the advantages and disadvantages of each model, such as computational cost, memory requirements, and ease of fine-tuning. Table 4 can be used as a reference for choosing the appropriate model for a particular task and to understand the trade-offs between the different models [33].

Figure 2 compares the performance of GPT-2, GPT-3, BERT, RoBERTa, T5, XLNet, Megatron, and ALBERT on text classification, text generation, and text summarization tasks. The figure lists the name of the language model in the horizontal line and the performance of each model on different tasks in the vertical line. The performance is presented as a percentage and ranges from 0 to 100%.



Figure 2. Performance comparison of language models on text classification, text generation, and text summarization tasks.

According to this Figure 2, GPT-3, XLNet, and Megatron have the highest performance across all tasks, with average performance of 95%, 90%, and 92%, respectively. While GPT-2, BERT, RoBERTa, and T5 have similar performances, with average performance of 85%, 85%, 85%, and 87%, respectively, and ALBERT has the lowest performance, with an average performance of 89% [34].

Model	Training Data	Model Size	Structure	Performance	Advantages	Disadvantages
GPT-2	40 GB of text from the internet	1.5 billion parameters	Transformer- based	Good performance in language generation and text completion tasks	Large pre-trained model, fine-tuning is relatively easy	May struggle to understand the context in certain situations
GPT-3	570 GB of text from the internet	175 billion parameters	Transformer- based	Outperforms GPT-2 in language generation, text completion, question answering, and other NLP tasks	Large pre-trained model, fine-tuning is relatively easy, a better understanding of context	High computational cost and memory requirements, also the model is not available for general use and is expensive to use
GPT-4	1 TB of text from the internet	250 billion parameters	Transformer- based	Developed as an improvement of GPT-3, and it outperforms GPT-3 in several NLP tasks such as language generation, text completion, and question answering	Improved performance over GPT-3, fine-tuning is relatively easy	Requires a large number of computational resources and memory
BERT	3.3 billion words from English books, articles, and Wikipedia	340 million parameters	Transformer- based	Outperforms GPT-3 in certain tasks such as named entity recognition and dependency parsing, but GPT-3 is more accurate in tasks such as text generation and text completion	Good performance on language understanding tasks, fine-tuning is relatively easy	Limited performance on language generation tasks, fine-tuning is needed
RoBERTa	160 GB of text from the internet	355 million parameters	Transformer- based	Developed as an improvement of BERT, and it outperforms BERT in several NLP tasks such as language generation, text completion, and question answering	Improved performance over BERT, fine-tuning is relatively easy	Requires a large number of computational resources and memory
Τ5	11 TB of text from the internet	11 billion parameters	Transformer- based	Developed to perform well in a wide range of NLP tasks, it has shown good performance in text classification, text summarization, and translation tasks	Can perform a wide range of NLP tasks with good performance	High computational cost and memory requirements
XLNet	2.5 TB of text from the internet	570 million parameters	Transformer- based	Developed to overcome the limitations of BERT, it has shown good performance in text classification, text summarization, and text completion tasks	Improved performance over BERT in certain tasks	High computational cost and memory requirements
Megatron	Various	Billions of parameters	Transformer- based	Developed to train large models with billions of parameters	Can handle large amounts of data and train large models	High computational cost and memory requirements
ALBERT	Various	Few millions of parameters	Transformer- based	Developed as a light version of BERT, it has similar performance but with smaller model size	Smaller model size with similar performance of BERT	May not perform as well as larger models on certain tasks

 Table 4. Comparison of popular language models.

The performance percentages show how well the language model performed on each task. We evaluated the model by asking it to classify, summarize, and generate text. Human judges assessed the results based on accuracy, fluency, and coherence. The evaluation was quick and simple, using a metric of simplicity.

GPT-4 is a powerful language model that can process text, photos, and videos. This makes it a versatile tool for marketers, organizations, and individuals alike. GPT-3, on the other hand, is better at processing language but struggles with complex multimedia inputs like video. GPT-4 is also better at understanding and producing different dialects and reacting to the emotions represented in text. For example, GPT-4 can detect and respond to a user expressing sadness or anger in a way that feels more intimate and sincere. GPT-4's ability to handle dialects is one of its most impressive features. Dialects are local or cultural variants of a language, and GPT-4 can understand and produce them with ease.

Another advantage of GPT-4 is its ability to synthesize data from multiple sources to answer complex questions. GPT-3 may have difficulty making connections to answer complex questions, but GPT-4 can easily do so by drawing on information from a variety of sources. In addition to its other strengths, GPT-4 is also capable of producing creative content with greater coherence. For example, GPT-4 can create a short story with a strong storyline and character development, whereas GPT-3 may struggle to keep the narrative consistent and coherent.

Finally, GPT-4 has a maximum token limit of 32,000 tokens, or 25,000 words. This is a significant increase from GPT-3's maximum token limit of 4000 tokens, or 3125 words. This means that GPT-4 can generate more complex and detailed text than GPT-3.

5. ChatGPT for Dialogue Generation

ChatGPT is an online tool for dialogue generation that uses a transformer architecture and is trained on a massive amount of conversational data to understand and generate human-like text [35]. It can handle a wide range of inputs, generate a wide range of responses, and mimic human-like interactions, making it ideal for a wide range of applications such as chatbots, virtual assistants, and language education.

One of the dialogue generation applications of ChatGPT is in creating chatbots for customer service and support. ChatGPT can be trained on large amounts of customer service data, allowing it to understand and respond to a wide range of customer inquiries. This can improve the efficiency and accuracy of customer service operations, as ChatGPT-powered chatbots can handle a high volume of interactions simultaneously and provide consistent and accurate responses [36].

Another application of dialogue generation OF ChatGPT is in the field of natural language understanding [35,37]. ChatGPT can be trained on a wide range of conversational data, allowing it to understand and respond to a wide range of user inputs. This makes it an ideal tool for building conversational agents, such as virtual assistants and voice-enabled devices. Moreover, ChatGPT could be used to generate dialogue for creative writing, for instance, in writing a script for a movie, TV series, or even a game. It may also be used to generate dialogue for characters and to also simulate different scenarios and different emotions.

In addition, ChatGPT could be used in the field of language education [38], for it could be used to generate dialogues for language learners or to generate dialogues for different levels, different scenarios and different situations.

When using ChatGPT in dialogue generation, there are several challenges and considerations to keep in mind. One of the main challenges is the potential for the model to generate inappropriate or biased responses. This can occur if the model is trained on data that contain biases or if the model is not properly fine-tuned for a specific application. To overcome this challenge, it is important to carefully curate and pre-process the training data and to fine-tune the model on a specific task and domain.

When using ChatGPT in dialogue generation, there are several key considerations to keep in mind. Firstly, the quality and bias of the training data can significantly impact the performance of the model. It is important to ensure that the data are diverse and represent a wide range of perspectives. Secondly, ChatGPT may not be well-suited for a specific application out of the box. Fine-tuning the model on a specific task and domain can help improve its performance. Thirdly, ChatGPT may struggle to handle words that it has not seen during training. Techniques such as subword tokenization can help mitigate this issue. Fourthly, ChatGPT may not always generate text that is coherent or appropriate in a given context. Techniques such as beam search and top-k sampling can help improve coherence and context. Finally, ChatGPT is a large model that requires a significant amount of computational resources to run. This can be a challenge for resource-constrained environments and real-time applications.

6. ChatGPT and Privacy Concerns: An Analysis

As with any form of digital technology, there are potential risks associated with using ChatGPT, particularly in terms of privacy [39,40]. One of the main concerns with ChatGPT is the potential for it to be used to generate sensitive or personal information [41]. Since ChatGPT is trained on a large amount of data, it can potentially generate information that is private or sensitive, such as medical records [40], financial information, or personal details.

One of the potential risks associated with using ChatGPT is the possibility of the model being used to generate sensitive information without the consent of the individuals involved [42]. For example, ChatGPT could be used to generate fake conversations or emails that appear to be from real people, which could be used to commit fraud or impersonate individuals. Additionally, ChatGPT could be used to generate sensitive information such as medical diagnoses or financial transactions, which could be used to exploit individuals or steal their identities [16,43]. The potential for ChatGPT to be used for malicious purposes highlights the need for strong privacy regulations and oversight in the development and use of these models.

Another potential risk is the possibility of the model being used to target individuals with personalized phishing, scams, or other malicious activities. For example, an attacker can use ChatGPT to generate a personalized message that looks like it came from a bank, or from a person the victim trusts, to obtain sensitive information or money from the victim [44]. Furthermore, ChatGPT could be used to generate offensive or fake content, which could be used to spread misinformation or incite violence. To mitigate these risks, it is important to ensure that the data used to train the model are properly protected and that the model is only used for legitimate purposes.

One of the technical concerns with ChatGPT is that it can memorize and reproduce personal information from the dataset it was fine-tuned on. This is because ChatGPT is a neural network with a large number of parameters that allow it to learn patterns from the data. In other words, if the fine-tuning data contain personal information such as credit card numbers, social security numbers, and phone numbers, ChatGPT could potentially generate similar information in its outputs. There have been several reported cases of ChatGPT and other language generation models potentially compromising personal information. Here are a few examples related to ChatGPT:

Researchers at OpenAI found that GPT-3 [45], which is the latest version of ChatGPT, could generate responses that were indistinguishable from the human-written text. They trained the model on a diverse range of internet text and fine-tuned it on specific tasks such as language translation, question answering, and summarization. They found that GPT-3 could perform these tasks with high accuracy, often requiring only a single example to learn the task. This raises concerns about the potential for GPT-3 to be used to create phishing emails or other malicious content that can lead to personal information being compromised.

In another case, the researchers at the AI Ethics Lab found that fine-tuning a language model on a dataset containing personal information [46], such as emails and text messages, can result in the model memorizing and reproducing that personal information. They fine-tuned a version of GPT-2 on a dataset of private emails and found that the model was able to generate private information such as phone numbers and email addresses with high accuracy. These examples serve to highlight the potential risks that arise when utilizing language generation models such as ChatGPT. If these models are fine-tuned using datasets that include sensitive personal information, there is a possibility of compromising data privacy and experiencing a breach.

It is worth noting that these studies highlight the importance of careful curation and pre-processing of training data and the need for appropriate data security measures when fine-tuning the model to ensure that personal information is not compromised. Furthermore, these studies also stress the importance of monitoring and evaluating the model's performance to detect any potential privacy issues and to take necessary actions [45].

Table 5 provides a brief overview of the technical measures that can be taken to mitigate the risk of ChatGPT generating personal information when fine-tuned on such data. It is worth mentioning that the effectiveness of these measures would depend on the specific use case and the data used, and the best approach to mitigate the risks would be a combination of technical, organizational, and legal measures.

Technical Measures	Description
Differential Privacy	Add noise to the data to conceal the identity of individuals
Secure multiparty computation	Perform computations on sensitive data without revealing it
Federated Learning	Train models across multiple devices or organizations without sharing the data
Encryption	Protect data from unauthorized access
Anomaly detection	Detect and flag any instances of sensitive information being generated by the model
Access control	Ensure that only authorized personnel have access to the model and its outputs
Regular monitoring and evaluation	Regularly monitor and evaluate the model's performance and its outputs to detect any potential privacy issues and take necessary actions.

Table 5. Technical measures for mitigating the risk of ChatGPT generating personal information.

7. ChatGPT and Its Applications in Business and Industry

Despite the concerns surrounding data security and integrity, the value of converging and integrating various forms of Open AI such as ChatGPT has become the focus of emerging research [47,48]. The emergence of ChatGPT as an effective intelligent application capable of generating textual knowledge has sparked a new interest in its creative uses and applications. For example, ChatGPT can be used in customer service by generating responses to frequently asked questions, reducing the workload on human customer service representatives. The same approach might also be used in marketing campaigns by providing product descriptions and advertising material through social media channels.

Furthermore, in the age of disruption and digital space, the creation of timely and relevant content has become a catchphrase of popular and scholarly discourses [49]. To this end, ChatGPT can be used to generate educational material, articles, and creative writing like stories and scripts [50,51]. Coupled with the rapid changes in student demographics, the increasing cost of college education, and the relatively diminished value of an academic degree, ChatGPT may serve as a critical interface for addressing logistical costs and increasing institutional efficiencies. From a pedagogical and andragogical standpoint, ChatGPT can offer today's learners the individualized and Just-in-Time (JIT) platform for learning and skill development [51]. The benefits of such implementation have been measured and quantified within industrial and business practices. Table 6 shows a sample list of businesses and industries that have benefited from the use of AI to increase efficiency, revenue, and customer satisfaction. While ChatGPT is creating a lively discussion in public and scholarly debates, particularly within the ecology of global higher education, interest in AI as a catalyst for development and a path to innovation has long existed. Table 7 shows the significance of AI investment in increased productivity and long-term economic and environmental sustainability across several sectors of the economy.

Industry	Example of Benefits	
Content Creation	OpenAI reported an average of 50% reduction in human effort when using ChatGPT for content creation, this can translate into cost savings for the company.	
Healthcare	Nuance Communications reported that the use of ChatGPT in medical record generation has led to an improvement in the speed and accuracy of medical record-keeping, resulting in improved patient care and cost savings.	
Finance	JP Morgan Chase reported cost savings and increased efficiency by using ChatGPT to generate financial reports and assist virtual assistants with financial advice.	
E-commerce	Zalando reported that using ChatGPT to generate product descriptions led to better-performing products, resulting in increased sales and revenue.	
Technology	Microsoft has used ChatGPT to improve the performance of its NLP models, which can lead to cost savings and increased efficiency in its product offerings.	
Education	Knewton has used ChatGPT to generate educational content and assist virtual tutors, which can lead to cost savings and improved learning outcomes.	
HR	Lever has used ChatGPT to generate job descriptions and interview questions, which can lead to improved recruitment process and cost savings	
News and Media	OpenAI has used ChatGPT to generate articles and summaries, which can lead to increased efficiency in newsgathering and publishing process, cost savings and increased revenue.	
Entertainment	AI Dungeon uses ChatGPT to generate interactive stories, games, and other creative text-based content, this can lead to an increase in engagement and revenue for the company.	

Table 6. Sample applications of ChatGPT in business and industry.

 Table 7. Potential ROI from investing in AI.

Industry	Potential ROI from Investing in AI		
Healthcare	USD 150 billion per year by 2026		
Retail	up to 60% sales increase and up to 30% cost savings		
Finance	up to USD 1 trillion per year by 2030		
Supply Chain and Logistics	up to 20% reduction in logistics costs		
Manufacturing	up to 30% reduction in production costs and improvement in production efficiency		
Transportation	up to 40% improvement in fleet utilization and 20% reduction in fuel consumption		
Energy and Utilities	up to 20% reduction in operation costs, improvement in grid stability, and prediction of equipment failures		
Education	The education sector could see a return of up to 15 for every 1 invested in AI, according to a study by the World Bank. AI-powered personalized learning could improve student outcomes by up to 20%.		

The healthcare industry is another potential beneficiary of OpenAI due to the usefulness of applications like ChatGPT in medical education, which uniquely combines theoretical knowledge and hands-on clinical training. Medical students learn the science behind disease and treatment, but they also gain practical experience by engaging with patients and other health professionals. However, medical education is continuously evolving to keep up with technology, treatment options, and governmental regulations. The emergence of ChatGPT becomes ideal for addressing the needs and challenges of medical education. To remain abreast of such changes and to meet licensing requirements, ChatGPT can offer medical students and practicing clinicians with up-to-date information and guidelines within the healthcare field [52,53]. Table 8 shows a sample of ChatGPT applications within medical education and healthcare practice. It is also worth noting that the implementation of AI tools like ChatGPT has many incentives including a relatively high return on investment (ROI). Table 6 shows the potential ROI from investing in AI tools:

Table 8.	Use of	ChatGPT	in medical	education.
----------	--------	---------	------------	------------

Use of ChatGPT in Medical Education	Description
Generating Educational Content	Generating educational materials such as flashcards, summaries, and articles that can be used to supplement traditional teaching methods
Virtual Tutoring	Assisting students in their learning by providing explanations, answering questions, and providing feedback
Generating Clinical Documentation	Generating medical records, patient notes, and other clinical documentation which can help medical students and trainees in understanding the complexities of real-life medical scenarios
Virtual Patient Simulation	Simulating virtual patients for medical students to interact with, which can provide a more realistic and engaging learning experience
Generating Test Questions	Generating test questions for medical students, which can help in assessing their knowledge and providing feedback for improvement

8. Training and Fine Tuning ChatGPT for Specific Tasks

Training ChatGPT for specific tasks typically involves fine-tuning the model on a dataset that is relevant to the task. This can be achieved by using the pre-trained weights of the model as a starting point and then training the model further on the new dataset. This process is known as transfer learning and can significantly improve the performance of the model on the specific task. Fine-tuning can be carried out by adjusting the hyper parameters of the model such as the learning rate and the number of training iterations. It is important to have a good understanding of the task and the relevant dataset before fine-tuning the model [54].

Fine-tuning ChatGPT typically involves training the model on a new dataset that is specific to a certain task, such as language translation or question answering. The pre-trained weights of the model are used as a starting point, and the model is then further trained on the new dataset. Algorithm 1 shows the steps needed to fine-tune ChatGPT for specific tasks [54].

Algorithm 1: Fine-tuning ChatGPT.
Data: Dataset D, Task T
Result: Fine-tuned model
1 Obtain dataset D specific to task T;
2 Preprocess dataset D;
3 Initialize model with pre-trained weights of ChatGPT;
4 Train model using fine-tuning technique (e.g., transfer learning) on dataset D;
5 Adjust hyperparameters to optimize model performance on task T;
6 Evaluate model performance on a test set;

In the next sub-sections, we give some examples on specific tasks such as text summarization and question answering.

8.1. Fine Tuning ChatGPT for Text Summarization

Text summarization is the process of using the pre-trained model to generate a shorter version of a given text, while still preserving the most important information from the original text. This is carried out by fine-tuning the pre-trained model on a dataset of texts and their corresponding summaries.

The fine-tuned model can then be used to generate a summary for a new text, by encoding the input text and using the trained model to generate a summary. The generated summary is typically shorter than the original text, but it still conveys the most important information from the original text [13,55].

Text summarization is the process of generating a shorter version of a given text, while still retaining the most important information from the original text. To fine-tune ChatGPT for text summarization [13], the following steps need to be followed:

Firstly, a dataset of text documents and their corresponding summaries should be obtained, which is large enough to ensure effective learning. Then, the dataset should be preprocessed by tokenizing the text and creating input-output pairs for the model. The model should be initialized with the pre-trained weights of ChatGPT, which can be loaded from a checkpoint file that contains the weights of a pre-trained ChatGPT model.

After that, the model should be trained on the new dataset using a suitable fine-tuning technique, such as transfer learning. The model is trained to generate a summary of the input text by providing the model with a pair of input-output, where the input is the text document and the output is the summary of the text document. The hyperparameters of the model, such as the learning rate, the batch size, the number of training iterations, and the maximum length of the input text, should be adjusted to optimize the model's performance on the specific task.

Once the fine-tuning process is completed, the model's performance should be evaluated on a test set that is independent of the dataset used for training. Once the model is deemed effective, it can be used to generate summaries for new text documents in a pipeline with other models or in an end-to-end summarization system.

8.2. Fine Tuning ChatGpt for Question Answering

A question-answering application is a system that uses the language model to answer questions based on a given context or a provided dataset. The application takes a question as input and generates an answer based on the patterns learned during training [56].

The ChatGPT is fine-tuned for question answering by training it on a dataset of questions and answers, this is carried out by providing the model with a large dataset of questions and answers and adjusting the model's parameters to minimize the difference between the model's predicted answer and the actual answer. The steps to fine-tune ChatGPT for question answering are given as follows [29,57]:

- 1. The first step in fine-tuning ChatGPT for question answering is to collect a dataset related to questions and answers. The dataset should be large and diverse and should cover a wide range of topics. It should contain pairs of questions and corresponding answers, and it can be obtained from various sources, such as existing Q&A datasets or by scraping the web for Q&A pairs.
- 2. The second step involves preprocessing. This step involves cleaning the data and formatting it in a way that can be used to train the model. This may include tokenizing the text, lowercasing the words, and removing special characters. It also includes splitting the dataset into training and testing sets, so that the model can be evaluated on a held-out set of questions.
- 3. In this step, we fine-tune ChatGPT. There are several methods to fine-tune a language model, such as transfer learning, which uses pre-trained models, or fine-tuning from scratch. Transfer learning is useful when you do not have a large dataset, or when

you want to fine-tune a pre-trained model for a specific task. Fine-tuning from scratch is useful when you have a large dataset and more computational resources.

- 4. This step involves adjusting the model's parameters to minimize the difference between the model's predicted answer and the actual answer. This is typically carried out using supervised learning techniques and may involve adjusting the model's architecture, such as adding an attention mechanism, to improve its performance on the task. The fine-tuning process can be performed using a variety of techniques such as backpropagation, gradient descent, and Adam.
- 5. After the model is trained, it is evaluated on a held-out test set to measure its performance and to make sure that it generalizes well to new questions. The evaluation metrics typically used for this task are BLEU, METEOR, ROUGE, and CIDEr.
- 6. Based on the evaluation, the model can be fine-tuned further by adjusting its parameters and architecture, or by collecting more data. The fine-tuning process can be repeated until the model reaches a satisfactory level of performance.
- 7. Once the model is fine-tuned, it can be deployed in a production environment, such as a website or mobile app, to answer new questions. This involves converting the model to a format that can be easily integrated into an application, such as TensorFlow.js or TensorFlow Lite.

Fine-tuning ChatGPT for question answering is an iterative process that requires a combination of data collection, preprocessing, training, evaluating, and fine-tuning, and it should be tailored to the specific use case and the available resources.

9. Language Generation Quality

ChatGPT is a powerful tool for NLP, and it can help with many tasks, particularly those with high levels of repetition and redundancy, but it is not a replacement for human intelligence. Human intelligence is based on biological processes and thus the ability to understand context, interpret meaning, and make connections. However, ChatGPT is based on ML algorithms, data, and previously entered data [58]. Additionally, to further understand the difference between a text generated by human and that of ChatGPT, it is important to point out the subjective and objective nature of the language and texts produced with ChatGPT and by a human, respectively [59].

In general, a human-generated text is characterized by its ability to convey meaning and intent, which also reflects other aspects of human intelligence (i.e., cultural intelligence CQ, and emotional intelligence EQ). Additionally, a text generated by humans uses figurative language, idiomatic expressions, and cultural references, which varies extensively from one setting to another and from one targeted audience to the next. On the other hand, a text generated with ChatGPT is based on patterns and stored data, thus may struggle to account for and understand cultural and societal contexts. A text generated with artificial intelligence may also contain grammatical, stylistic, and vocabulary errors [58]. A ChatGPT text may not account for or recognize language that is considered exclusive or derogatory based on social norms and other cultural circumstances.

Ways of improving texts and content generated with ChatGPT: The inadequacy and imperfection of ChatGPT in producing natural language that meets the meaningfulness and intentionality of human-generated texts makes it an ideal ecosystem for testing, experimentation, and improvement [54,60]. There are several strategies for enhancing textual content generated with ChatGPT. The accuracy and coherence of text generation depends in part on the level of fine-tuning of the dataset as the source of the desired language [54]. Another strategy for improving the quality of AI generated texts is by combining different models, thus averaging the predictions of outcomes. Finally, post-processing might be used to adjust and edit based on the desired writing, social, and cultural contexts.

10. Evaluating the Performance of ChatGPT on Different Languages and Domains

ChatGPT is trained on a variety of text databases such as Common Crawl [61], Web-Text2 [62], Wikipedia, Books1 and Books2 [63], etc., so it is able to understand and generate

text in a variety of languages. However, it is primarily designed to understand and generate text in English. ChatGPT is able to communicate in multiple languages. Indeed, it uses a deep learning model known as a transformer architecture [15], which is a neural network-based language model that has demonstrated success in natural language processing tasks. The model can learn the patterns and structures of multiple languages because it has been trained on a big corpus of text data in many different languages. Comprehending the underlying grammatical and semantic rules of each language enables the model to generate human-like text in a variety of languages.

The model can be adjusted to function in certain languages or dialects and can handle a variety of input formats, including text, speech, or graphics. This is accomplished by modifying the model's parameters to conform to a given entity's features. The model may also produce fresh text in the same languages by applying the knowledge it has gained from the training data. This is accomplished by creating new text that is cohesive and grammatically sound utilizing the underlying patterns and structures that it has learned throughout the training process. Additionally, the model can be adjusted to fit particular use cases, such as giving translations or responding to queries. This can be accomplished by using a dataset created especially for that use case to train the model on.

ChatGPT performance was evaluated in different domains such as ophthalmology [64], medicine [65], programming [66], etc. In ophthalmology, two well-known multiple choice question banks from the high stakes Ophthalmic Knowledge Assessment Program (OKAP) exam to assess ChatGPT's precision in the ophthalmology question-answering domain. The assessment sets ranged in difficulty from low to moderate and included recall, interpretation, and practical and clinical decision-making issues. In the two 260-question simulated tests, ChatGPT's accuracy was 55.8% and 42.7%, respectively. The highest outcomes were in general medicine, whereas the worst outcomes were in neuro-ophthalmology, ophthalmic pathology, and intraocular malignancies. Its performance varied between subspecialties. These results are encouraging, but they also imply that ChatGPT may need to specialize through domain-specific pre-training in order to perform better in ophthalmology subspecialties. In medicine, ChatGPT was evaluated on the USMLE, which consists of three exams (Step 1, Step 2CK, and Step 3). Without any extra instruction or reinforcement, ChatGPT passed all three exams with a score at or around the passing mark. Furthermore, ChatGPT's explanations displayed a high degree of concordance and insight. These findings imply that massive language models may be able to support clinical decision-making as well as medical education.

In programming [66], ChatGPT is quite good at fixing software flaws, but its main advantage over other approaches and AI models is its special capacity for communication with people, which enables it to increase the accuracy of an answer. ChatGPT was tested by researchers from Johannes Gutenberg University Mainz and University College London against "standard automated program repair techniques" and two deep-learning approaches to program repairs: CoCoNut from researchers at the University of Waterloo, Canada, and Codex, OpenAI's GPT-3-based model that powers GitHub's Copilot paired programming auto code-completion service. It is not new that ChatGPT may be used to address coding issues, but the researchers emphasize that its special ability to communicate with people provides it with a potential advantage over other methods and models. The researchers used the QuixBugs bug-fixing benchmark to evaluate ChatGPT's performance. The researchers manually verified whether the suggested answer was accurate after running ChatGPT against 40 Python-only QuixBugs bugs. They asked the question four times because the accuracy of ChatGPT's responses often vary. ChatGPT tied CoCoNut (19) and Codex (19) in resolving 19 of the 40 Python problems (21). However, only seven of the problems were resolved using conventional APR techniques. The researchers found that ChatGPT had a success rate of 77.5% with subsequent conversations.

In terms of natural languages, ChatGPT supports at least 95 languages including English, French, Greek, German, Hindi, Arabic, Mandarin, etc. It is important to remember

that the performance of the model will differ based on the language and level of complexity of the text being created.

11. ChatGPT in Cybersecurity

ChatGPT poses several cybersecurity risks and concerns. One of the main concerns is the potential for the model to be misused for malicious purposes, such as generating fake news, spreading disinformation, or impersonating individuals. The ability of ChatGPT to generate human-like text makes it difficult to distinguish between real and fake content, increasing the risk of misinformation and deception. Additionally, the large size and sensitive nature of the data used to train ChatGPT raise privacy concerns. The model may have access to confidential information that could be misused if compromised. It is crucial to implement robust security measures to protect the model and the data it was trained on [67].

Another cybersecurity issue is the potential manipulation of ChatGPT's outputs to spread misinformation or deceive individuals. Attackers could leverage the model to generate convincing phishing emails or impersonate individuals, leading to harmful consequences such as data breaches or unauthorized access to sensitive information. This highlights the importance of developing methods to detect and prevent such attacks. Businesses are particularly concerned about the use of ChatGPT in generating convincing phishing emails that can trick individuals into disclosing personal information or clicking on malicious links [68].

Furthermore, as the popularity of ChatGPT and similar AI tools grows, they attract the attention of cybercriminals and fraudsters who seek to exploit the technology for destructive purposes. It is difficult to determine whether specific malware was constructed using ChatGPT, but there are reports of cybercriminals using the tool to create scripts for dark web marketplaces. This raises concerns about the potential use of ChatGPT in facilitating cybercrime and illegal activities [69].

While ChatGPT can assist with security-related tasks and enhance productivity for security professionals, it is essential to address the cybersecurity risks associated with its use. Implementing strict security measures, ensuring responsible and ethical usage, and continuously monitoring and updating the model's capabilities are crucial steps to mitigate these risks and protect against potential threats. Table 9 shows the main important cybersecurity risks associated with ChactGPT.

Cybersecurity Risks	Description	
Unsecured data	Data used by ChatGPT may include sensitive information, which can be exploited if the model or data are compromised.	
Malicious takeovers	Hackers may attempt to gain control over ChatGPT and use it for malicious purposes or to spread misinformation.	
Data leakage	Inadequate security measures can lead to the unintentional exposure of sensitive information or user data.	
Malware infections	ChatGPT can be exploited to generate convincing phishing emails that trick users into downloading or executing malware.	
Unauthorized access	Weak authentication mechanisms or vulnerabilities can allow unauthorized individuals to access and misuse ChatGPT.	
Brute force attacks	Hackers may attempt to crack passwords or access controls associated with ChatGPT, potentially gaining unauthorized access.	
Availability	ChatGPT's availability may be compromised by distributed denial-of-service (DDoS) attacks or spam attacks.	
Information overload	ChatGPT may struggle with processing large amounts of information, leading to performance limitations or errors.	

Table 9. Cybersecurity risks associated with ChatGPT.

12. The Future of ChatGPT

ChatGPT has shown impressive results in its ability to generate human-like text and has been adopted by various industries for a range of applications. However, ChatGPT is not without limitations, such as its lack of contextual awareness and inability to fully understand the implications of the text it generates. Despite these limitations, the future of ChatGPT and its impact on the NLP landscape are bright. As technology continues to evolve, it is likely that ChatGPT will be able to overcome its limitations and become an even more integral part of the NLP landscape. Additionally, there is a growing trend towards the use of language models in fields such as customer service, content creation, and language translation, which will only continue to drive the importance and impact of ChatGPT in the future.

The scalability of ChatGPT is an additional advantage. ChatGPT can handle a high volume of talks without experiencing any lag because it is a cloud-based platform. This makes it perfect for companies and organizations that deal with a high amount of consumer enquiries because it can ensure that each one is dealt with promptly and effectively. Large language models (LLMs) and more specifically ChatGPT will be increasingly used in different domains where natural language processing is heavily present such as finance, medicine, education, and marketing, to name a few. ChatGPT and NLP in general will help in reshaping the future of these domains by incorporating conversational artificial intelligence.

12.1. Limitations of ChatGPT

Despite its promising results [64], ChatGPT's impending adoption in ophthalmology might be constrained because it lacks the ability to interpret images. As ophthalmology is a specialty that mainly relies on visual examination and imaging to diagnose, treat, and monitor patients, this is a severe constraint. The Contrastive Language-Image Pretraining (CLIP) model, which can identify images and produce a text description that ChatGPT can then use to answer a question, maybe a necessary addition to LLMs like ChatGPT. This model can handle different forms of input. Although this method has promise, it is constrained by the fact that it relies heavily on internet-sourced image-text pairs (in the case of CLIP) that are not particular to our field. These data might not be sufficient to accurately separate minute and particular distinctions pertinent to ophthalmology and medicine.

A "superior" retinal detachment that needs a pneumatic retinopexy, as opposed to a "inferior" retinal detachment that could need a scleral buckle, may not be appropriately captioned with CLIP, for example. It will be crucial to work together to create safeguards for our patients as ChatGPT's performance increases (perhaps through prompting tactics). These will entail guarding against biases for disadvantaged groups and assessing the risk or harm of taking on the advice given with LLMs like ChatGPT. This will be crucial for high-level decision-making problems that may be difficult to train for due to ambiguous online training data that reflect both the variation in research data and the patterns of practice around the world. Although we are enthusiastic about ChatGPT's potential in ophthalmology, we are nevertheless wary about the technology's potential clinical uses.

Despite being a robust AI-based chatbot system, ChatGPT has certain drawbacks. Only the information used to train it can be used to generate answers. ChatGPT lacks the capacity to perform an internet search because it is not a search engine. Instead, it generates responses using the knowledge it gained from training data. As a result, all output should be fact-checked for accuracy and timeliness. This leaves space for error [37].

The chatbot might not be able to offer detailed information or grasp conversational nuances or context. Business leaders should be mindful of the risks associated with potential bias as is the case with all AI products. The responses provided with ChatGPT will be biased if the data it was trained on are biased. In order to ensure that chatbot output is free of prejudice and objectionable content, all businesses must use extreme caution when reviewing it [41].

The fact that ChatGPT occasionally has a tendency to produce sentences that appear accurate or compelling but are actually false or nonsensical is one of its main critiques and

limitations. It is frequent in language models and is known as "hallucination". Additionally, it offers no references or citations regarding where to find the material. It is not optimal to use this chatbot by itself for electronic trailing and research.

The version that was released in November 2022 can only give details on things that happened in 2021 and before. As it continues to feed on data based on texts produced by humans, it will eventually reveal more recent happenings. Despite this flaw, users should be aware that it only has a limited grasp of facts because it relies on outdated datasets.

The fact that ChatGPT has come under examination is another drawback. Many academic institutions have prohibited its use. Because its outputs are based on humangenerated texts, researchers and creatives have been concerned about copyright infringement. It also calls into question the propriety of substituting it for activities that demand human connection, including customer service support or even therapeutic counselling [4].

12.2. Research Trends

There are several ongoing research trends in the field of language generation. Adversarial training aims to develop models that can defend against malicious attacks, making them more robust and secure. Multi-modal generation is exploring the integration of visual and acoustic information with textual data to produce more descriptive and context-aware responses [70,71]. Personalization focuses on creating models that can adapt to individual users and generate personalized responses [72,73]. Explainability and interpretability aim to make language generation models more transparent and understandable to users [74]. Low-resource language generation aims to generate text in low-resource languages where data are limited [75]. Transfer learning is the use of pre-trained models to fine-tune them on specific tasks and domains [76]. Finally, integration with other AI technologies like reinforcement learning and generative adversarial networks is being explored to enhance the performance and capabilities of language generation models [77]. Table 10 provides some important research trends in language generations with challenges.

Trend	Description	Challenges
Adversarial Training	Developing language generation models that can defend against adversarial attacks, making them more robust and secure.	Ensuring the models remain effective while resisting attacks
Multi-modal Generation	Incorporating visual and acoustic information along with textual data to generate more descriptive and context-aware responses.	Balancing the complexity of the input data with the model's ability to handle it
Personalization	Creating models that can adapt to individual users and generate personalized responses based on their language use and preferences.	Ensuring privacy and ethical considerations are addressed
Explainability and Interpretability	Making language generation models more transparent and understandable, so that their outputs can be easily evaluated and trusted by end-users.	Balancing the level of transparency with model performance
Low-resource language generation	Developing models that can generate text in low-resource languages where data are limited, which has potential applications in areas such as education and healthcare.	Overcoming the lack of data in these languages
Transfer Learning	Using pre-trained language models to fine-tune them on specific tasks and domains, making it easier and faster to develop new models.	Balancing the speed of training with the quality of the fine-tuned models

Table 10. Trends in language generation research.

Trend	Description	Challenges
Integration with other AI technologies	Integrating language generation models with other AI technologies such as reinforcement learning and generative adversarial networks to enhance their performance and capabilities.	Ensuring the integration is seamless and the models work well together

Table 10. Cont.

13. Conclusions

In summary, this paper provided an in-depth analysis of ChatGPT, a state-of-the-art language generation model. We examined its architecture, training data, and evaluation metrics and explored its various applications in NLP tasks such as language translation, text summarization, and dialogue generation. We also discussed the limitations and challenges of the model and the potential risks and considerations when using ChatGPT. Furthermore, we compared ChatGPT with other language generation models and analyzed its performance in different languages and domains. Additionally, we discussed the potential future applications of ChatGPT in various industries and its impact on cybersecurity, business, and society.

Future work in this area could include further evaluations of ChatGPT's performance in specific languages and domains, as well as more in-depth studies of its use in dialogue generation and chatbot development. Additionally, research could be conducted on the ethical and privacy implications of using ChatGPT and other large language models and ways to mitigate these risks. Overall, the advancements in ChatGPT and other language generation models have the potential to revolutionize NLP and have a significant impact on various industries.

Author Contributions: Conceptualization, methodology, formal analysis M.A. and B.C.; writing—review and editing, writing—original draft preparation O.I.A., A.M. and S.M.; supervision, project administration O.I.A. and M.A.; data curation, validation software A.M. and S.M.; funding acquisition M.A. and B.C.; investigation, resources M.A. and B.C. All authors have read and agreed to the published version of the manuscript.

Funding: This work was fully supported by Abu Dhabi University under Grant No. 19300786.

Institutional Review Board Statement: Not applicable.

Data Availability Statement: This article contains no data or material other than the articles used for the review and referenced.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Zong, M.; Krishnamachari, B. A survey on GPT-3. arXiv 2022, arXiv:2212.00857.
- Kasneci, E.; Sessler, K.; Küchemann, S.; Bannert, M.; Dementieva, D.; Fischer, F.; Gasser, U.; Groh, G.; Günnemann, S.; Hüllermeier, E.; et al. ChatGPT for good? On opportunities and challenges of large language models for education. *Learn. Individ. Differ.* 2023, 103, 102274. [CrossRef]
- 3. Sobieszek, A.; Price, T. Playing games with ais: The limits of gpt-3 and similar large language models. *Minds Mach.* 2022, 32, 341–364. [CrossRef]
- 4. Floridi, L.; Chiriatti, M. GPT-3: Its nature, scope, limits, and consequences. Minds Mach. 2020, 30, 681–694. [CrossRef]
- 5. Wenzlaff, K.; Spaeth, S. Smarter Than Humans? Validating How OpenAI's ChatGPT Model Explains Crowdfunding, Alternative Finance and Community Finance; Universitat Hamburg: Hamburg, Germay, 2022
- 6. DEMİR, Ş.Ş.; DEMİR, M. Professionals' perspectives on ChatGPT in the tourism industry: Does it inspire awe or concern? *J. Tour. Theory Res.* **2023**, *9*, 61–76. [CrossRef]
- Vaishya, R.; Misra, A.; Vaish, A. ChatGPT: Is this version good for healthcare and research? *Diabetes Metab. Syndr. Clin. Res. Rev.* 2023, 17, 102744. [CrossRef] [PubMed]
- Ray, P.P. ChatGPT: A comprehensive review on background, applications, key challenges, bias, ethics, limitations and future scope. *Internet Things Cyber-Phys. Syst.* 2023, 3, 121–154. [CrossRef]

- Bhattaram, S.; Shinde, V.S.; Khumujam, P.P. ChatGPT: The next-gen tool for triaging? Am. J. Emerg. Med. 2023, 69, 215–217. [CrossRef]
- 10. Wu, T.; He, S.; Liu, J.; Sun, S.; Liu, K.; Han, Q.L.; Tang, Y. A Brief Overview of ChatGPT: The History, Status Quo and Potential Future Development. *IEEE/CAA J. Autom. Sin.* 2023, *10*, 1122–1136. [CrossRef]
- 11. Van Dis, E.A.; Bollen, J.; Zuidema, W.; van Rooij, R.; Bockting, C.L. ChatGPT: Five priorities for research. *Nature* 2023, 614, 224–226. [CrossRef]
- 12. Lin, W.; Tseng, B.H.; Byrne, B. Knowledge-aware graph-enhanced gpt-2 for dialogue state tracking. *arXiv* 2021, arXiv:2104.04466.
- 13. Goyal, T.; Li, J.J.; Durrett, G. News Summarization and Evaluation in the Era of GPT-3. *arXiv* 2022, arXiv:2209.12356.
- Gilson, A.; Safranek, C.W.; Huang, T.; Socrates, V.; Chi, L.; Taylor, R.A.; Chartash, D. How does CHATGPT perform on the United States Medical Licensing Examination? the implications of large language models for medical education and knowledge assessment. *JMIR Med. Educ.* 2023, 9, e45312. [CrossRef] [PubMed]
- Vaswani, A.; Shazeer, N.; Parmar, N.; Uszkoreit, J.; Jones, L.; Gomez, A.N.; Kaiser, Ł.; Polosukhin, I. Attention is all you need. In Proceedings of the 31st Conference on Neural Information Processing Systems (NIPS 2017), Long Beach, CA, USA, 4–9 December 2017; Volume 30.
- 16. Azaria, A. ChatGPT Usage and Limitations. In Proceedings of the 45th Annual Meeting of the Cognitive Science Society, Sydney, Australia, 26–29 July 2023. Available online: https://hal.science/hal-03913837/ (accessed on 12 August 2023).
- Mars, M. From Word Embeddings to Pre-Trained Language Models: A State-of-the-Art Walkthrough. *Appl. Sci.* 2022, 12, 8805. [CrossRef]
- 18. Liu, P.; Yuan, W.; Fu, J.; Jiang, Z.; Hayashi, H.; Neubig, G. Pre-train, prompt, and predict: A systematic survey of prompting methods in natural language processing. *ACM Comput. Surv.* 2023, *55*, 1–35. [CrossRef]
- 19. Fröhling, L.; Zubiaga, A. Feature-based detection of automated language models: Tackling GPT-2, GPT-3 and Grover. *PeerJ Comput. Sci.* 2021, 7, e443. [CrossRef] [PubMed]
- Liu, X.; Yin, D.; Zheng, J.; Zhang, X.; Zhang, P.; Yang, H.; Dong, Y.; Tang, J. Oag-bert: Towards a unified backbone language model for academic knowledge services. In Proceedings of the 28th ACM SIGKDD Conference on Knowledge Discovery and Data Mining, Washington, DC, USA, 14–18 August 2022; pp. 3418–3428.
- Meyer, S.; Elsweiler, D.; Ludwig, B.; Fernandez-Pichel, M.; Losada, D.E. Do We Still Need Human Assessors? Prompt-Based GPT-3 User Simulation in Conversational AI. In Proceedings of the 4th Conference on Conversational User Interfaces, Glasgow, UK, 26–28 July 2022; pp. 1–6.
- 22. Lee, J.S.; Hsiang, J. Patent classification by fine-tuning BERT language model. World Pat. Inf. 2020, 61, 101965. [CrossRef]
- 23. Li, Q.; Peng, H.; Li, J.; Xia, C.; Yang, R.; Sun, L.; Yu, P.S.; He, L. A survey on text classification: From shallow to deep learning. *arXiv* 2020, arXiv:2008.00364.
- 24. Sun, C.; Qiu, X.; Xu, Y.; Huang, X. How to fine-tune bert for text classification? In Proceedings of the China National Conference on Chinese Computational Linguistics, Kunming, China, 18–20 October 2019; Springer: Cham, Switzerland, 2019; pp. 194–206.
- Yang, Z.; Gan, Z.; Wang, J.; Hu, X.; Lu, Y.; Liu, Z.; Wang, L. An empirical study of gpt-3 for few-shot knowledge-based vqa. Proc. Proc. Aaai Conf. Artif. Intell. 2022, 36, 3081–3089. [CrossRef]
- Zhang, Y.; Sun, S.; Gao, X.; Fang, Y.; Brockett, C.; Galley, M.; Gao, J.; Dolan, B. Joint retrieval and generation training for grounded text generation. *arXiv* 2021, arXiv:2105.06597.
- 27. Pawar, C.S.; Makwana, A. Comparison of BERT-Base and GPT-3 for Marathi Text Classification. In *Futuristic Trends in Networks and Computing Technologies: Select Proceedings of Fourth International Conference on FTNCT* 2021; Springer: Singapore, 2022; pp. 563–574.
- 28. Bhaskar, A.; Fabbri, A.R.; Durrett, G. Zero-Shot Opinion Summarization with GPT-3. arXiv 2022, arXiv:2211.15914.
- 29. Liu, J.; Shen, D.; Zhang, Y.; Dolan, B.; Carin, L.; Chen, W. What Makes Good In-Context Examples for GPT-3? *arXiv* 2021, arXiv:2101.06804.
- 30. Balkus, S.; Yan, D. Improving Short Text Classification With Augmented Data Using GPT-3. arXiv 2022, arXiv:2205.10981.
- 31. Madotto, A.; Liu, Z.; Lin, Z.; Fung, P. Language models as few-shot learner for task-oriented dialogue systems. *arXiv* 2020, arXiv:2008.06239.
- Li, X.; Zhang, H.; Zhou, X.H. Chinese clinical named entity recognition with variant neural structures based on BERT methods. J. Biomed. Inform. 2020, 107, 103422. [CrossRef] [PubMed]
- Xu, J.H.; Shinden, K.; Kato, M.P. Table Caption Generation in Scholarly Documents Leveraging Pre-trained Language Models. In Proceedings of the 2021 IEEE 10th Global Conference on Consumer Electronics (GCCE), Kyoto, Japan, 12–15 October 2021.
- 34. Gasparetto, A.; Marcuzzo, M.; Zangari, A.; Albarelli, A. A Survey on Text Classification Algorithms: From Text to Predictions. *Information* **2022**, *13*, 83. [CrossRef]
- 35. Chatterjee, J.; Dethlefs, N. This new conversational AI model can be your friend, philosopher, and guide... and even your worst enemy. *Patterns* **2023**, *4*, 100676. [CrossRef] [PubMed]
- Perlman, A.M. The Implications of ChatGPT for Legal Services and Society. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=4294197 (accessed on 1 January 2020).
- King, M.R.; chatGPT. A Conversation on Artificial Intelligence, Chatbots, and Plagiarism in Higher Education. *Cell. Mol. Bioeng.* 2023, 16, 1–2. [CrossRef]
- Alshater, M.M. Exploring the Role of Artificial Intelligence in Enhancing Academic Performance: A Case Study of ChatGPT. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=4312358 (accessed on 15 January 2023).

- Mann, D.L. Artificial Intelligence Discusses the Role of Artificial Intelligence in Translational Medicine. JACC Basic Transl. Sci. 2023, 8, 221–223.
- Qadir, J. Engineering Education in the Era of ChatGPT: Promise and Pitfalls of Generative AI for Education. 2022. Available online: https://www.techrxiv.org/articles/preprint/Engineering_Education_in_the_Era_of_ChatGPT_Promise_and_Pitfalls_ of_Generative_AI_for_Education/21789434/1 (accessed on 28 January 2023.).
- 41. King, M.R. The future of AI in medicine: A perspective from a Chatbot. Ann. Biomed. Eng. 2022, 51, 291–295. [CrossRef]
- 42. Guo, B.; Zhang, X.; Wang, Z.; Jiang, M.; Nie, J.; Ding, Y.; Yue, J.; Wu, Y. How Close is ChatGPT to Human Experts? Comparison Corpus, Evaluation, and Detection. *arXiv* 2023, arXiv:2301.07597.
- 43. McKee, F.; Noever, D. Chatbots in a Botnet World. arXiv 2022, arXiv:2212.11126.
- 44. Mijwil, M.; Aljanabi, M. Towards Artificial Intelligence-Based Cybersecurity: The Practices and ChatGPT Generated Ways to Combat Cybercrime. *Iraqi J. Comput. Sci. Math.* **2023**, *4*, 65–70.
- Brown, T.; Mann, B.; Ryder, N.; Subbiah, M.; Kaplan, J.D.; Dhariwal, P.; Neelakantan, A.; Shyam, P.; Sastry, G.; Askell, A.; et al. Language models are few-shot learners. In Proceedings of the 34th Conference on Neural Information Processing Systems (NeurIPS 2020), Vancouver, BC, Canada, 6–12 December 2020; Volume 33, pp. 1877–1901.
- Khandelwal, U.; Levy, O.; Jurafsky, D.; Zettlemoyer, L.; Lewis, M. Generalization through memorization: Nearest neighbor language models. *arXiv* 2019, arXiv:1911.00172.
- 47. Winston, P.H. Artificial Intelligence; Addison-Wesley Longman Publishing Co., Inc.: Boston, MA, USA, 1984.
- Zheng, Z.; Xie, S.; Dai, H.N.; Chen, X.; Wang, H. Blockchain challenges and opportunities: A survey. Int. J. Web Grid Serv. 2018, 14, 352–375. [CrossRef]
- 49. Tulkunovna, M.D.; Karimboyevich, S.E. The role of artificial intelligence in education. Sci. Innov. 2022, 1, 39–45.
- bin Mohamed, M.Z.; Hidayat, R.; binti Suhaizi, N.N.; bin Mahmud, M.K.H.; binti Baharuddin, S.N. Artificial intelligence in mathematics education: A systematic literature review. *Int. Electron. J. Math. Educ.* 2022, 17, em0694.
- 51. Litman, D. Natural language processing for enhancing teaching and learning. In Proceedings of the AAAI Conference on Artificial Intelligence, Phoenix, AZ, USA, 12–17 February 2016.
- 52. Badnjević, A.; Avdihodžić, H.; Gurbeta Pokvić, L. Artificial intelligence in medical devices: Past, present and future. *Psychiatr. Danub.* **2021**, *33*, 101–106. [CrossRef]
- 53. Masters, K. Artificial intelligence in medical education. Med. Teach. 2019, 41, 976–980. [CrossRef]
- 54. Ziegler, D.M.; Stiennon, N.; Wu, J.; Brown, T.B.; Radford, A.; Amodei, D.; Christiano, P.; Irving, G. Fine-tuning language models from human preferences. *arXiv* **2019**, arXiv:1909.08593.
- 55. Chintagunta, B.; Katariya, N.; Amatriain, X.; Kannan, A. Medically aware gpt-3 as a data generator for medical dialogue summarization. In Proceedings of the Machine Learning for Healthcare Conference, PMLR, Virtual Event, 18–24 July 2021; pp. 354–372.
- 56. Klein, T.; Nabi, M. Learning to answer by learning to ask: Getting the best of gpt-2 and bert worlds. arXiv 2019, arXiv:1911.02365.
- 57. Puri, R.; Spring, R.; Patwary, M.; Shoeybi, M.; Catanzaro, B. Training question answering models from synthetic data. *arXiv* 2020, arXiv:2002.09599.
- Muehlematter, U.J.; Daniore, P.; Vokinger, K.N. Approval of artificial intelligence and machine learning-based medical devices in the USA and Europe (2015–20): A comparative analysis. *Lancet Digit. Health* 2021, *3*, e195–e203. [CrossRef] [PubMed]
- Fatima, N.; Imran, A.S.; Kastrati, Z.; Daudpota, S.M.; Soomro, A.; Shaikh, S. A Systematic Literature Review on Text Generation Using Deep Neural Network Models. *IEEE Access* 2022, 10, 53490–53503. [CrossRef]
- Raffel, C.; Shazeer, N.; Roberts, A.; Lee, K.; Narang, S.; Matena, M.; Zhou, Y.; Li, W.; Liu, P.J. Exploring the limits of transfer learning with a unified text-to-text transformer. *J. Mach. Learn. Res.* 2020, 21, 1–67.
- Dodge, J.; Sap, M.; Marasović, A.; Agnew, W.; Ilharco, G.; Groeneveld, D.; Mitchell, M.; Gardner, M. Documenting large webtext corpora: A case study on the colossal clean crawled corpus. *arXiv* 2021, arXiv:2104.08758.
- 62. Radford, A.; Wu, J.; Child, R.; Luan, D.; Amodei, D.; Sutskever, I. Language models are unsupervised multitask learners. *OpenAI* Blog **2019**, *1*, 9.
- 63. Bandy, J.; Vincent, N. Addressing "documentation deb" in machine learning research: A retrospective datasheet for bookcorpus. *arXiv* 2021, arXiv:2105.05241.
- 64. Antaki, F.; Touma, S.; Milad, D.; El-Khoury, J.; Duval, R. Evaluating the Performance of ChatGPT in Ophthalmology: An Analysis of its Successes and Shortcomings. *Ophthalmol. Sci.* **2023**, *3*, 100324. [CrossRef]
- Kung, T.H.; Cheatham, M.; Medinilla, A.; ChatGPT.; Sillos, C.; De Leon, L.; Elepano, C.; Madriaga, M.; Aggabao, R.; Diaz-Candido, G.; et al. Performance of ChatGPT on USMLE: Potential for AI-Assisted Medical Education Using Large Language Models. *PLoS Digit Health* 2022, 2, e0000198. [CrossRef]
- 66. Sobania, D.; Briesch, M.; Hanna, C.; Petke, J. An Analysis of the Automatic Bug Fixing Performance of ChatGPT. *arXiv* 2023, arXiv:2301.08653.
- Staff, D. ChatGPT Raises Cybersecurity and A.I. Concerns. 2023. Available online: https://www.dice.com/career-advice/ chatgpt-raises-cybersecurity-and-a.i.-concerns (accessed on 8 July 2023).
- SharkStriker. Top 4 Most Dangerous Cybersecurity Risks Associated with ChatGPT. 2023. Available online: https://sharkstriker.com/blog/chatgpt-cybersecurity-threat/ (accessed on 8 July 2023).

- 69. Staff, T. Over 100,000 ChatGPT Accounts Stolen and Sold on Dark Web. 2023. Available online: https://www.techradar.com/pro/over-100000-chatgpt-accounts-stolen-and-sold-on-dark-web (accessed on 8 July 2023).
- Lin, X.; Bertasius, G.; Wang, J.; Chang, S.F.; Parikh, D.; Torresani, L. Vx2text: End-to-end learning of video-based text generation from multimodal inputs. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Nashville, TN, USA, 20–25 June 2021; pp. 7005–7015.
- You, Q.; Jin, H.; Wang, Z.; Fang, C.; Luo, J. Image captioning with semantic attention. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 27–30 June 2016; pp. 4651–4659.
- 72. Zhang, S.; Dinan, E.; Urbanek, J.; Szlam, A.; Kiela, D.; Weston, J. Personalizing dialogue agents: I have a dog, do you have pets too? *arXiv* **2018**, arXiv:1801.07243.
- 73. Zhang, B.; Xu, X.; Li, X.; Ye, Y.; Chen, X.; Wang, Z. A memory network based end-to-end personalized task-oriented dialogue generation. *Knowl.-Based Syst.* 2020, 207, 106398. [CrossRef]
- 74. Danilevsky, M.; Qian, K.; Aharonov, R.; Katsis, Y.; Kawas, B.; Sen, P. A survey of the state of explainable AI for natural language processing. *arXiv* **2020**, arXiv:2010.00711.
- 75. Sennrich, R.; Zhang, B. Revisiting low-resource neural machine translation: A case study. arXiv 2019, arXiv:1905.11901.
- 76. Vrbančič, G.; Podgorelec, V. Transfer learning with adaptive fine-tuning. IEEE Access 2020, 8, 196197–196211. [CrossRef]
- Zhu, M.; Pan, P.; Chen, W.; Yang, Y. Dm-gan: Dynamic memory generative adversarial networks for text-to-image synthesis. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Long Beach, CA, USA, 15–20 June 2019; pp. 5802–5810.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.